Author's response to reviews

Title: High-dose antioxidants for central serous chorioretinopathy; The randomized, placebo-controlled study.

Authors:

Mansing Ratanasukon (mratanasukon@yahoo.com)
Patama Bhurayanontachai (patama103@yahoo.com.au)
Pichai Jirarattanasopa (jpichai@hotmail.com)

Version: 5 Date: 13 June 2012

Author's response to reviews: see over
13 June, 2012

Dear Editor,

Enclosed please find the revised manuscript, "High-dose antioxidants for central serous chorioretinopathy; The randomized, placebo-controlled study", for re-consideration for publication in "BMC Ophthalmology". This study has not been published elsewhere.

The research was approved by our Ethics Committee, Faculty of Medicine, Prince of Songkla University on December 2004 (EC 47/362-023), and before the enrollment, all patients signed the consent forms. The research has also registered in the www.ClinicalTrials.gov (NCT 00963131). We did not have financial support from any industrial company for this study.

We also included the final responses to the reviewer’ comment as follow,

The responses to reviewer comment (Michael Marmor)
Comment ; The reviewer suggested a final English correction on 2 sentences on page 12.
Response & answer; The page-12 sentences were corrected (as underlined), followed the comments.

Finally, we did the manuscript checklist one more time, followed the journal guidelines.

Sincerely yours,

Mansing Ratanasukon, MD
Associate Professor
Department of Ophthalmology, Faculty of Medicine
Prince of Songkla University
Hat Yai, Songkhla Province
Thailand 90110
Tell: 66 - 74 -451381
Fax: 66 - 74 - 429619
E-mail address: mratanasukon@yahoo.com